

Answering Reviewers:

Reviewer 02623966: It is an interesting manuscript. Authors succeed to present their data in a clear way adding information to the existing literature. Therefore, I have no corrections to do and the manuscript can be published unaltered.

Answering: Thank you for your kind response.

Issues raised 1:

The authors did not provide the approved grant application form(s). Please upload the approved grant application form(s) or funding agency copy of any approval document(s); The study was supported by Guangzhou Science and Technology Plan. The topic has not previously been published in the WJCC.

The authors provided the written informed consent, however, the informed consent of treatment was not provided.

Answering: We have uploaded the approved grant document(s). We also checked carefully the funding topic and have revised the translation of the funding agency, the topic should be “Guangzhou Science and Technology Program key projects”.

Additionally, due to the government action against the COVID-19, all relevant infected individuals and close contacts were forced to treat and isolate observation, respectively. Hence we can not provide the informed consent of treatment for we did not collect it as usual.

Issues raised 2:

The authors did not provide original pictures. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor;

Answering: We have provided original figure documents with the version of PowerPoint.

Issues raised 3:

PMID and DOI numbers are missing in the reference list. Please provide the PubMed numbers and DOI citation numbers to the reference list and list all authors of the references. Please revise throughout;

Answering: We checked carefully and have provided the PubMed numbers and DOI citation

numbers to the reference list and list all authors of the references.

Issues raised 4:

The “Case Presentation” section was not written according to the Guidelines for Manuscript Preparation. Please re-write the “Case Presentation” section, and add the “FINAL DIAGNOSIS”, “TREATMENT”, and “OUTCOME AND FOLLOW-UP” sections to the main text, according to the Guidelines and Requirements for Manuscript Revision;

Answering: We checked carefully and have re-written the “Case Presentation” section according to the Guidelines for Manuscript Preparation.

Issues raised 5:

Please add the all authors’ names with institution information, ORCID and emails to the system.

Answering: We have added all authors’ names with institution information, ORCID and emails to the system.

Issues raised 6:

Company editor-in-chief:the quality of the English language of the manuscript does not meet the requirements of the journal. Before final acceptance, the author(s) must provide the English Language Certificate issued by a professional English language editing company. Please visit the following website for the professional English language editing companies we recommend: <https://www.wjgnet.com/bpg/gerinfo/240>.

Answering: We visit the website <https://www.wjgnet.com/bpg/gerinfo/240> for the professional English language editing companies and have provided the locked paper with Certificate.